Author’s response to reviews

Title: Dietary patterns in Canadian Men and Women Ages 25 and Older: Relationship to Demographics, Body Mass Index, and Bone Mineral Density

Authors:

Lisa Langsetmo (lisa.langsetmo@sympatico.ca)
Suzette Poliquin (suzette.poliquin@mail.mcgill.ca)
David A Hanley (dahanley@ucalgary.ca)
Jerilynn C Prior (Jerilynn.Prior@vch.ca)
Susan Barr (susan.barr@ubc.ca)
Tassos Anastassiades (anastass@post.queensu.ca)
Tanveer Towheed (tt5@post.queensu.ca)
David Goltzman (david.goltzman@mcgill.ca)
Nancy Kreiger (Nancy.Kreiger@cancercare.on.ca)
Canadian Multicentre Osteoporosis Study (CaMos) (info@camos.org)

Version: 4 Date: 18 January 2010

Author’s response to reviews:

Dear Colleagues:

All of the pre-publication changes requested by the editors have been made. These were submitted for me by the first author, Lisa Langsetmo, as I was travelling and had limited computer access.

Please let me know if there are any editorial request with which we have not complied, and I will make the corrections.

Thank you.

Sincerely,

David Hanley